

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Bijsterbosch, Martin; Kolasa, Marcin

Working Paper FDI and productivity convergence in central and eastern Europe: an industry-level investigation

ECB Working Paper, No. 992

Provided in Cooperation with: European Central Bank (ECB)

Suggested Citation: Bijsterbosch, Martin; Kolasa, Marcin (2009) : FDI and productivity convergence in central and eastern Europe: an industry-level investigation, ECB Working Paper, No. 992, European Central Bank (ECB), Frankfurt a. M.

This Version is available at: https://hdl.handle.net/10419/153426

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



WORKING PAPER SERIES NO 992 / JANUARY 2009

B EZB EKT EKP

FDI AND PRODUCTIVITY CONVERGENCE IN CENTRAL AND EASTERN EUROPE AN INDUSTRY-LEVEL

by Martin Bijsterbosch and Marcin Kolasa

INVESTIGATION





NO 992 / JANUARY 2009

FDI AND PRODUCTIVITY CONVERGENCE IN CENTRAL AND EASTERN EUROPE

AN INDUSTRY-LEVEL INVESTIGATION'

by Martin Bijsterbosch² and Marcin Kolasa³



This paper can be downloaded without charge from http://www.ecb.europa.eu or from the Social Science Research Network electronic library at http://srn.com/abstract_id=1311396.

In 2009 all ECB publications feature a motif taken from the €200 banknote.





 This paper was written while Marcin Kolasa was working in DG Economics of the ECB. The authors would like to thank the participants to the internal ECB seminar and the INFER Workshop in Cluj-Napoca for useful comments. Special thanks are owed to: Hans-Joachim Klöckers, Reiner Martin, Monica Pop-Silaghi and two anonymous referees. The views expressed in this paper are solely the responsibility of the authors and should not be interpreted as reflecting the views of the European Central Bank and the National Bank of Poland.
 2 European Central Bank, DG Economics, Kaiserstrasse 29, D-60311 Frankfurt am Main, Germany; e-mail: martin.bijsterbosch@ecb.europa.eu

3 National Bank of Poland, Economic Institute, ul. Swietokrzyska 11/21, 00-919 Warsaw, Poland; e-mail: Marcin.Kolasa@mail.nbp.pl

© European Central Bank, 2009

Address Kaiserstrasse 29 60311 Frankfurt am Main, Germany

Postfach 16 03 19 60066 Frankfurt am Main, Germany

Telephone +49 69 1344 0

Website http://www.ecb.europa.eu

Fax +49 69 1344 6000

All rights reserved.

Any reproduction, publication and reprint in the form of a different publication, whether printed or produced electronically, in whole or in part, is permitted only with the explicit written authorisation of the ECB or the author(s).

The views expressed in this paper do not necessarily reflect those of the European Central Bank.

The statement of purpose for the ECB Working Paper Series is available from the ECB website, http://www.ecb.europa. eu/pub/scientific/wps/date/html/index. en.html

ISSN 1725-2806 (online)

CONTENTS

Abstract						
No	n-teo	chnical summary	5			
1	Intr	oduction	7			
2	2 Stylized facts on productivity convergence and FDI inflows in central and eastern Europe					
3	The	oretical considerations	13			
4	Eco	nometric strategy	15			
5 Data sources and definitions of variables						
6	6 Results					
	6.1	Preliminary regressions	20			
	6.2	FDI and absorptive capacity	21			
	6.3	The role of openness and innovation	24			
	6.4	Cross-section heterogeneity	24			
	6.5	Two periods of convergence	26			
	6.6	Some robustness checks	27			
7	Con	clusions	28			
Re	ferer	ices	29			
Ch	arts	and tables	33			
Annex						
European Central Bank Working Paper Series						

3

Abstract

This paper presents empirical evidence of the effect of FDI inflows on productivity convergence in central and eastern Europe, using industry-level data. Four conclusions stand out. First, there is a strong convergence effect in productivity, both at the country and at the industry level. Second, FDI inflow plays an important role in accounting for productivity growth. Third, the impact of FDI on productivity critically depends on the absorptive capacity of recipient countries and industries. Fourth, there is important heterogeneity across countries, industries and time with respect to some of the main findings.

Keywords: productivity convergence, FDI, absorptive capacity

JEL classification: C23, F21, O33

Non-technical summary

The central and eastern European EU Member States have recorded impressive productivity gains over the past 15 years. Despite this catching-up process, however, a marked gap vis-à-vis the rest of the EU remains. These productivity gains have been accompanied by substantial inflows of FDI, which have been facilitated by supportive government policies. These capital inflows are generally considered to be the main vehicle for economic restructuring and technology diffusion. The empirical evidence on the link between productivity and FDI in the central and eastern European region has, however, been more mixed, probably due to a lack of cross-country and crossindustry data.

A key question is thus how important FDI inflows have been for the convergence process in general and for productivity gains in particular. If FDI has a consistent positive impact on productivity, this would imply that countries should continue to pursue policies aimed at attracting FDI. In addition, it is important to understand whether and which economic conditions affect the size of the benefits associated with FDI inflows. Studies on productivity growth have underlined the importance of absorptive capacity, and tacit knowledge in particular, which may enhance the transfer of technology and thereby strengthen the impact of FDI on productivity growth. The emphasis on absorptive capacity is based on the idea that the potentially positive impact of FDI on the receiving economy may fail to materialise if domestic companies lack sufficient abilities to imitate and adopt superior technologies used by foreign firms. This capacity to absorb technology depends on a wide range of factors, such as levels of basic technological literacy and advanced skills or on the quality of the business environment in general. Absorptive capacity can be measured in various ways, for example on the basis of human capital indicators or using the relative productivity level.

This paper provides empirical evidence of the overall effect of FDI inflows for productivity convergence in central and eastern Europe, using industry-level data from a relatively new and to a large extent still unexploited database (EU KLEMS). These data have a country, industry and time dimension, covering a wide range of countries and sectors in a consistent way. An important feature of the paper is that it also concentrates on whether the size of benefits associated with FDI depends on the absorptive capacity of the recipient country. The robustness of the empirical results in the paper is checked by relying on two alternative econometric approaches, one exploiting the cross section while the other the time dimension of the data.

The results in this paper point to the following conclusions. First, there is a strong convergence effect in productivity both at the country and at the industry level, i.e. productivity growth depends positively on its gap vis-à-vis the euro area. At the country level, this effect is highly pronounced in the Baltic region. At the industry level, the convergence effect is particularly strong in the manufacturing sector. Second, foreign capital, in the form of FDI inflows, plays an important role in accounting for productivity growth in the central and eastern European region. Third, the impact of FDI on productivity critically depends on the capacity to absorb technology. More specifically, the effect of FDI on productivity seems to be increasing with a declining productivity differential vis-à-vis the euro area. There is also evidence that the level of human capital is positively associated with a larger impact of FDI. The former type of interaction between absorptive capacity and the beneficial impact from FDI seems to be important in manufacturing, whereas the latter is more significant in services.

Overall, using a new harmonised industry-level database, this paper provides empirical evidence that FDI and absorptive capacity are key factors for productivity convergence in central and eastern Europe. The policy implication of this result is that creating favourable conditions for FDI is likely to support productivity convergence. More importantly, however, the favourable impact of FDI on productivity is not automatic and can be strengthened by improving the absorptive capacity of the recipient economy, for example via raising the level of human capital.

Working Paper Series No 992 January 2009

1. Introduction

After more than 15 years of transition and despite an impressive catching-up process, productivity levels in central and eastern European EU countries remain substantially below those in the rest of the EU. In 2005, for example, average GDP per capita in the region stood at around 55% of the euro area (see Arratibel et al., 2007).¹ Further raising productivity levels, therefore, remains a key priority for economic policies in these countries. Understanding developments in productivity is also crucial for the analysis of short-term dynamics of GDP growth and for imbalances between supply and demand, as productivity is a key determinant of the supply side of the economy.

The catching-up process in central and eastern Europe has coincided with large inflows of foreign direct investment (FDI). A key question arising from this phenomenon is how important FDI inflows have been for the convergence process in general and for productivity gains in particular. If FDI has a consistent positive impact on productivity, this would imply that countries should continue to pursue policies aimed at attracting FDI. In addition, it is important to understand whether and which economic conditions affect the size of the benefits associated with FDI inflows.

The existing cross-country studies on growth and productivity in central and eastern Europe largely have a stocktaking or growth-accounting character and concentrate mostly on the macro-level (see e.g. Campos and Coricelli, 2002; Doyle et al., 2001; European Commission, 2004; Lenain and Rawdanowicz, 2004). The main focus of this literature is on the pace and nature of the growth process, concentrating on the period since the start of the transition to a market economy. These studies mostly underline the importance of economic policies (including institutions) for growth and convergence.

There have been a number of attempts to investigate the link between FDI and economic growth in a more formal way. Only very few of them, however, take a cross-country or cross-industry perspective, mainly due to the lack of comparable data. For instance, Holland and Pain (1998) examine the early stages of transition in central and eastern Europe (1992-1996). They estimate a labour demand function using aggregate data for eight countries and find that the stock of inward foreign

¹ Weighted average of the Czech Republic, Estonia, Hungary, Latvia, Lithuania, Poland, Slovakia and Slovenia. In this paper, central and eastern Europe refers to these eight EU countries.

investment has a positive impact on productivity, with the beneficial effects being higher in the more market-orientated economies. These results were broadly confirmed by a related study of Barrell and Holland (2000), based on industry-level data covering eleven manufacturing sectors in the Czech Republic, Hungary and Poland.

Other empirical evidence on the role of FDI in the catching-up process of transition economies has been less conclusive. For example, Mencinger (2003) applies a Granger causality test to aggregate data covering eight central and eastern European economies in the period 1994-2001. The main finding is that the relationship between FDI and GDP growth is negative, which is attributed to the character of FDI during this period (mostly privatisation-related rather than greenfield investment). Hunya (1997) estimates that foreign-owned enterprises operating in the region have on average higher labour productivity than domestic firms, but notes that this may be related to the concentration of the former in more capital intensive industries. This hypothesis was confirmed by Djankov and Hoekman (2000), who use firm-level data from the Czech Republic and conclude that, after controlling for various kinds of selection biases, FDI does not seem to have a significant effect on productivity growth.

Following an improved availability of firm-level data, an increasing number of papers on the link between FDI and growth have focused on productivity spillovers from foreign-owned companies to other firms in the economy.² A key conclusion emanating from these studies is that spillovers vary by country, sector and type of firm, depending also on the nature of FDI and absorptive capacity of domestic enterprises.

The approach taken by the most recent spillover literature has many advantages over the earlier studies. While it is certainly of interest whether there are any externalities associated with FDI inflows, an important question remains about their total impact on aggregate productivity. More specifically, since multinational companies are among the most technologically advanced firms, their presence may be beneficial for a recipient country even if their superior knowledge does not spill over

 $^{^2}$ See a meta-analysis by Görg and Greenaway (2004) or an investigation in a cross-country setup by Damijan et al. (2003). More recent contributions to the spillover literature focusing on central and eastern Europe include Gersl et al. (2007), Gorodnichenko et al. (2007) and Kolasa (2008).

to domestic firms. Therefore, it may still be useful to take a less disaggregated perspective for assessing the macroeconomic impact of FDI.

A relatively detailed industry-level analysis has recently become possible thanks to the EU KLEMS database. It covers a wide range of sectors in an internationally comparable way and can be considered as a state-of-the-art source for cross-country and cross-industry comparisons. With this relatively new and to a large extent still unexploited database, we can carry out an econometric analysis, using data with both a country, industry and time dimension. To the best of our knowledge, this has not been done for transition economies before.

There are two other important features of our paper that distinguish it from the previous literature. First, we investigate formally how the size of overall benefits associated with FDI depends on the absorptive capacity of the recipient country, which allows us to shed more light on the cross-country variation in the estimated effects of FDI found in previous studies. Second, in the empirical part we employ two alternative econometric approaches, differing in the extent to which they exploit the industry versus the time dimension of the data. This makes our main conclusions more robust compared to earlier studies, relying only on one single method.

The rest of the paper is structured as follows. Section 2 provides stylized facts on the catching-up process and foreign capital inflow to central and eastern Europe. In Section 3, we discuss some theoretical considerations underlying our empirical investigation. Section 4 presents the econometric strategy. Section 5 describes data sources and definitions of variables. The main results and robustness checks are discussed in Section 6. Section 7 concludes.

2. Stylized facts on productivity convergence and FDI inflows in central and eastern Europe

Despite impressive increases over the past 15 years, labour productivity levels in central and eastern Europe remain well below those in the rest of the EU. Taking the euro area as a benchmark, Charts 1 and 2 provide an overview of relative labour productivity (calculated as value added per hour worked, see Section 5 for more details).³

Chart 1 shows that there is substantial heterogeneity in productivity levels across sectors. Whereas productivity gaps vis-à-vis the euro area in the mid-1990s were relatively large in industry, they were substantially less pronounced in construction and market services. In particular, output per hour in financial and business-related services was relatively close to that in the euro area already in the mid-1990s. Despite marked increases over the past decade, labour productivity levels in almost all sectors remain consistently lower than in the euro area. This applies particularly to industry, where output per hour was on average only around one third of the euro area in 2006. In services, productivity levels were on average around half of those in the euro area, whereas in construction the pattern was quite diverse across countries.

[Chart 1 and 2 about here]

Focusing on the dynamics since 1995, Chart 2 shows that cumulative productivity gains in industry since the mid-1990s have outpaced those in the other sectors. Whereas there seems to be a strong convergence effect present in industry, such a pattern was less clearly visible in construction and services. Within the service sector, however, the patterns are not homogenous. In the more traditional services, most countries in the group managed to raise productivity significantly, with cumulative gains between 1995 and 2006 mostly ranging from 20% to 70% (apart from Estonia, where productivity increases were much higher, see the charts in the Annex for a country-by-country overview). In the financial and business-related services, by contrast, productivity gains remained more limited, particularly in the central European countries (the Czech Republic, Hungary, Poland, Slovakia and

³ For presentational reasons, the individual industries for which data are available have been lumped together in this section into four broadly defined sectors. Industry, in the first panel, mainly consists of manufacturing, together with mining and quarrying and electricity, water and gas supply (NACE categories D, C and E, respectively). The second sector is construction (NACE F). The third and fourth sectors are (market) services, with the former covering the more "traditional" services, such as trade and repairs, hotels and restaurants as well as transport and communication (NACE G, H and I), while the latter comprises financial and business-related services (NACE J and K). These four sectors together cover all economic activities except agriculture (and related branches) and non-market services.

Slovenia). The industrial sector thus seems to be the main driver of labour productivity convergence vis-à-vis the euro area. This productivity pattern across sectors seems intuitive, given the differences in capital intensity and technology content between these broadly defined sectors.

A similar convergence effect seems to be present at the macro-level across the countries considered, as the economies with the lowest initial productivity levels have been catching up relatively rapidly. This applies in particular to the Baltic countries (Estonia, Latvia and Lithuania), where labour productivity levels increased from around a quarter of those in the euro area in the mid-1990s to around 30-40% in 2006. In the central European countries, where output per hour was on average around 40% of the euro area level in 1995, productivity rose to around 50% of the euro area in 2006.

Productivity convergence in central and eastern Europe has often been associated with FDI inflows, which are considered to be the main vehicle for economic restructuring and technology diffusion (see, for example, EBRD, 1994 or Damijan and Rojec, 2007). Central and eastern European countries have been quite successful in attracting FDI, also relative to other emerging market economies (see Castejón and Wörz, 2006). Annual changes in FDI stocks have averaged around 5% of GDP in the eight central and eastern European countries considered in this paper, though there were large fluctuations from year to year. Looking at the allocation of FDI across countries, Chart 3 shows that Estonia stood out in receiving the largest inflows, with the FDI stock increasing from around a quarter to almost 100% of GDP between 1997 and 2005. The Czech Republic and Hungary also recorded sizeable cumulated inflows and the FDI stock to GDP ratio was slightly above 50% in both economies in 2005. Overall, there does not seem to be a clear geographical pattern in FDI inflows.

[Chart 3 about here]

As regards the allocation of FDI across sectors, most inflows have gone to financial and business-related services and industry. Chart 4 shows that in these sectors FDI stocks relative to value added increased substantially over the past decade. The country-by-country charts in the Annex show that the high FDI intensity in Estonia, the Czech Republic and Hungary seems to be broad-based, with these

countries consistently having the highest FDI to value added ratios in industry and in both broadly defined service sectors.

At a more disaggregated level, by far the largest recipient of FDI in services was financial intermediation, followed by business-related services (i.e. real estate, renting and business activities) and trade. The FDI stock to value added ratio in the transport, storage and communication also increased very strongly, but reached a peak already around the turn of the century (whereas the FDI intensity in financial and business-related services exhibits a consistent upward trend). Although initially FDI in services seems to have been associated with privatisation, other motives like market seeking, cost reduction and (more recently) outsourcing seem to have been important driving factors as well (Gersl et al., 2007). Within industry, FDI inflows were concentrated in transport equipment, food, as well as electrical and optical equipment. FDI in industry seems to have been mainly motivated by cost reduction, although privation also played a role in the earlier FDI inflows.

[Chart 4 about here]

To conclude, this overview of the data can be summarised by three observations. First, the initial level of productivity matters for the subsequent speed of convergence towards the euro area. At the sectoral level, this convergence effect is illustrated by the relatively strong productivity increases in industry over the past decade following relatively low levels in the mid-1990s. At the macro-level, the relatively strong increases in output per hour in the Baltic States, which had relatively low productivity levels in the mid-1990s, point into the same direction. Second, FDI inflows have mostly been concentrated in financial and business-related services and, to a lesser extent, in industry, although FDI patterns have been rather diverse across sectors and countries. At the country level, Estonia, the Czech Republic and Hungary have been the main FDI recipients over the past decade relative to their economic size. Finally, considerable differences exist across countries and sectors both as regards productivity developments and FDI inflows, particularly at a more disaggregated level.

3. Theoretical considerations

While FDI is definitely not the only channel through which international technological diffusion may occur, it is widely considered to be the most important one. This is because multinational corporations are among the most technologically advanced firms, spending relatively big amounts on research and development and using better managerial practices. This implies that inward FDI may involve the transfer of superior technologies, which can then spread over the entire economy leading to productivity gains in domestic firms (see e.g. Findlay, 1978 or Romer, 1993).⁴

The link between inward FDI and economic growth in developing countries has firm theoretical foundations. As demonstrated by Borensztein et al. (1998), this relationship can be derived using the framework of international technology diffusion developed by Barro and Sala-i-Martin (1997) and drawing on seminal contributions to the theory of endogenous growth by Romer (1990) or Grossman and Helpman (1991).

According to this setup, per-capita (or labour productivity) growth occurs via accumulation of human capital and the expansion in the number of varieties of capital goods used in production of final goods. These varieties are produced by domestic and foreign firms that have undertaken a direct investment in the economy. An increase in the number of capital varieties requires a fixed cost of adapting the technology available in more advanced economies. This cost decreases with the share of foreign firms operating in the host economy and is negatively related to the technological gap vis-à-vis developed countries, which reflects decreasing imitation possibilities over the catching-up process.

Similar ideas can also be incorporated into neoclassical growth models. This was done e.g. by Wang (1990), who assumes that the increase in effective knowledge applied to production can be written as a function of FDI. A description of technology diffusion involving decreasing imitation possibilities during the convergence process with an important role of human capital is owed to Nelson and Phelps (1966). Duczynski (2003) incorporates the concept of international technology diffusion into a Ramsey framework with capital mobility and discusses the implications of his model in the context of transition economies.

⁴ In transition economies FDI inflows may also play an important role in the process of restructuring of formerly state-owned companies (see e.g. Blanchard, 1997).

On the basis of these theoretical considerations, one can write a simple model of productivity growth in a catching-up economy using inward FDI, the relative productivity level vis-à-vis developed economies and human capital as the main explanatory variables. The precise specification of the model can take different forms (see Section 4).

However, while FDI is generally considered to be a key channel for economic restructuring and international technology diffusion, FDI inflows as such may not necessarily be sufficient to ensure an increase in productivity. The extent to which these flows are translated into technological progress and productivity growth depends on the absorptive capacity of the sector and the country. This, in turn, hinges on the levels of basic technological literacy as well as on more advanced skills in the host country or sector (see e.g. World Bank, 2008).

The absorptive capacity concept can be implemented empirically by extending the simple model sketched out above to include interactions between the main explanatory variables. For instance, by interacting FDI with the relative productivity level we can examine to what extent gains from foreign capital inflows depend on the absorptive capacity, measured as the distance to the technological frontier. In particular, we could test the hypothesis of e.g. Glass and Saggi (1998),⁵ according to which a larger development gap implies a lower quality of technology transferred via FDI and more limited capabilities of domestic firms to benefit from potential spillovers of foreign presence (implying a negative coefficient of the interaction term). On the other hand, a positive estimate would be consistent with an alternative hypothesis provided by Findlay (1978), who emphasises the larger pool of available technological opportunities and a stronger pressure for change in relatively backward economies.

Absorptive capacity considerations can also be taken into account by interacting human capital with both the relative productivity level (used as a proxy for potential technology transfer) and FDI inflow. A classical reference stressing the role of human capital in technological diffusion is Nelson and Phelps (1966), who interact measures of human capital quality with the productivity gap vis-à-vis the technological frontier in their growth regressions.⁶ Borensztein et al. (1998) and

⁵ See also Kokko (1994).

⁶ See also Benhabib and Spiegel (2005). A confirmation of the Nelson-Phelps hypothesis, using a panel of OECD countries, can be found in Griffith et al. (2004).

Balasubramanyan et al. (1999) confirm the link between the impact of FDI and the quality of human capital.

It has to be noted that human capital and relative productivity are not the only proxies for absorptive capacity advocated in the literature. In particular, local firms' capabilities to absorb knowledge from abroad can be dependent on their own innovation effort (see Cohen and Levinthal, 1989). Also, a wide set of other characteristics (like competitive pressure, financial market development, regulations) can affect the speed of the catching-up process and the size of potential spillovers from FDI.

4. Econometric strategy

Having defined a set of potential explanatory variables, the choice of an appropriate econometric strategy is far from straightforward. In general, the most popular approaches followed in the empirical growth literature can be classified into two groups, which we will refer to as cross section and time series studies.

The first group comprises a vast literature exploiting mainly cross-country or (less frequently) cross-industry correlation between growth and a wide set of explanatory variables. The variables used in regressions are averaged over relatively long time spans covering the whole sample (e.g. Barro, 1991; Mankiw, Romer and Weil, 1992) or form a set of non-overlapping averages (e.g. Borensztein et al., 1998; Schadler et al., 2006).

The main advantage of cross section studies is that their results are less likely to be driven by cyclical movements. Moreover, by exploiting cross sectional information, they are potentially better suited for addressing questions about the sources of differences in performance across countries or industries. In practice, however, the latter advantage may be undermined by the omitted variables problem and endogeneity, leading to potentially serious biases in the estimates of the coefficients of interest.⁷

⁷ In principle, this kind of problems can be mitigated by using instrumental variable techniques. However, lack of good instruments makes this option rather impractical or can even do more harm than good (see Nelson and Startz, 1990 or Bound et al., 1995).

The second group of approaches, time series studies, aims at testing relationships of interest within rather than across countries or industries. This type of approach relies mainly on yearly observations and uses panel-data methods (see e.g. Islam, 1995; Griffith et al., 2004; Carkovic and Levine, 2005).

The biggest advantage of the time series approach is that it is less vulnerable to the sources of biases that may affect purely cross section regressions. This is because the inclusion of fixed effects in the panel helps to control for unobservable heterogeneity between objects considered, making the omitted bias problem less severe. Additionally, more sophisticated panel data techniques that rely on generalized method of moments (GMM) attempt to address the endogeneity issue, although in a rather mechanistic fashion. The major weakness of the time series approach is, however, that it does not exploit cross-section variation in the data and that it may not fully account for medium and long-run relationships by using data of relatively high frequency. Although there seems to be a tendency in the empirical growth literature towards using the time series approach, one has to bear in mind that it has its flaws.

Therefore, as a matter of robustness, it might be useful to check whether the results obtained using the other approach are at least qualitatively similar. Any striking discrepancy between the time series and cross section evidence would then call for caution in interpreting the results. Given the above considerations, our empirical investigation will rely on both approaches, the details of which are summarized below.

In the time series approach, we employ the system GMM estimation framework developed by Arellano and Bond (1991) and then extended by Arellano and Bover (1995) and Blundell and Bond (1998).⁸ More specifically, we regress the annual growth rate in labour productivity on the set of explanatory variables lagged one period, with a full set of time dummies.

The use of the system GMM method is motivated by the fact that our specification can be rewritten so that the level of productivity in central and eastern European Member States is expressed as a function of its own lag and the lagged level of productivity in the euro area. The presence of the lagged dependent variable implies that standard methods used for estimating panel data models, like the fixed

⁸ We use the xtabond2 procedure for Stata. See Roodman (2006).

effects estimator, produce biased results if the number of time periods in the sample is small (see Nickell, 1981). Lagging other explanatory variables, and FDI in particular, is aimed at avoiding a simultaneity bias,⁹ while including time dummies is expected to capture possible cyclical movements between productivity growth and right hand side variables, common across countries and industries.

The cross section approach is pursued by splitting the sample into two fiveyear periods and applying the SUR technique to a system of two equations.¹⁰ All variables in these equations are expressed as five-year averages, except for the relative productivity level, which is measured in the year preceding the beginning of the relevant period. It has to be noted that our sample is different from standard cross section studies in that it has both a country and an industry dimension, which makes it possible to do the estimations with a full set of country and industry dummies. Naturally, this is not equivalent to fully controlling for unobservable heterogeneity across objects (like in the time series approach with a full set of country-industry specific effects). However, it is reasonable to expect that this strategy will at least attenuate the possible bias afflicting traditional cross section estimations. An additional advantage of including country dummies is the fact that they can be regarded as (imperfect) substitutes for country-wide indicators usually used in the empirical growth literature (quality of institutions, size of the government, macroeconomic stability, financial market development etc.).

5. Data sources and definitions of variables

The main data source of which this paper makes use is the new EU KLEMS database. It is the result of a project carried out by a consortium of research institutes and financed by the European Commission in order to facilitate productivity analyses in the EU at the industry level (see Timmer et al., 2007).¹¹ The main adjustments to the

⁹ We treat all lagged explanatory variables as predetermined, which means that they are assumed to be uncorrelated with present and future errors. This assumption might be violated e.g. if FDI inflow is motivated by expectations of future shocks, which seems rather unlikely.

¹⁰ This means that our cross section approach also exploits some time series variation in the data, although to a much lesser extent than the system GMM technique applied to yearly data.

¹¹ EU KLEMS stands for EU analysis of capital (K), labour (L), energy (E), materials (M) and service (S) inputs. The database is downloadable at <u>www.euklems.net</u>. It consists of two types of variables: analytical (growth accounting) variables, such as labour and capital input, total factor productivity etc.,

official statistical sources made in the database relate to filling gaps in industry-level data (using industry statistics) and to linking series over time.¹²

A key advantage of the EU KLEMS database is that it covers a wide range of industries (up to 72 per country, including a breakdown of services) in an internationally comparable way, with the key variables anchored in official statistics. The database covers the EU Member States in central and eastern Europe from 1995 onwards. In addition, it includes a large number of variables that are potentially relevant for understanding productivity developments. These features make the database probably the state-of-the-art source for cross-country and cross-industry comparisons.

Despite the above advantages, it should be emphasised that the EU KLEMS database is still work in progress. The level of detail varies across countries, industries and variables, with some gaps in particular for the EU Member States in central and eastern Europe (see Table 1). In addition, the quality of the data is still being evaluated by the national statistical institutes of the countries concerned. More generally, it needs to be kept in mind that the measurement of productivity in services is surrounded by a number of conceptual and empirical caveats, which suggests that the data for these industries should be used with some degree of caution (this applies, however, not only to the EU KLEMS database).

Another data source used in the paper is the WIIW database on Foreign Direct Investment in Central, East and Southeast Europe (see Hunya and Schwarzhappel, 2007). It contains industry-level FDI data as reported by the national central banks of the countries in the region. A key advantage of the database is that the industry breakdown is consistent with the one in the EU KLEMS database. In addition, the data are harmonised in the sense that they are in line with standard IMF definitions and methodological guidelines (although some methodological changes over time have taken place). The FDI data in this paper come from the May 2007 release of the WIIW database.

and statistical variables, which are largely based on national accounts (ESA95) data of the individual countries. This paper only makes use of the latter group of EU KLEMS data as the coverage of the former for the EU Member States in central and eastern Europe is still insufficient for the purposes of this study. See also Koszerek at al. (2007) for an extensive overview of the database.

¹² These adjustments were done by the EU KLEMS consortium on the basis of agreed procedures to ensure harmonisation of the data and to generate growth accounts in a consistent and uniform way. Harmonisation focused, among others, on industrial classifications, aggregation levels, reference years for volume measures, price concepts and methods for solving breaks.

[Table 1 about here]

Table 1 provides an overview of the variables used in this study. Our total sample covers nineteen sectors of eight central and eastern European EU Member States and spans the period 1995-2005. The countries considered are: the Czech Republic, Estonia, Hungary, Latvia, Lithuania, Poland, Slovakia and Slovenia.¹³ The sectors covered are: manufacturing (13 industries: NACE DA to DN, without DC), construction (NACE F) and market services (5 industries: NACE G to K).¹⁴

We measure labour productivity (*LP*) as real value added per hour worked.¹⁵ Relative productivity levels (*RLP*) are calculated vis-à-vis the euro area using industry-specific purchasing power parities (PPPs) for 1997. Relative levels for the remaining years are extrapolated using labour productivity growth rates. The data are taken from the March 2007 release of the EU KLEMS database, covering the period 1995-2004.¹⁶

Our *FDI* variable is defined as the ratio of FDI inflow to gross value added. FDI inflow is calculated as a change in inward FDI stocks.¹⁷ The time span covered in the WIIW database differs across countries and (to lesser extent) across industries. Most FDI data for Latvia, Lithuania and Slovenia cover the whole period of interest (1995-2005), while one or two first years are missing for the other countries.

Human capital (*HC*) is measured as the share of hours worked by high-skilled persons in total hours worked. The share of high-skilled workers is available for all but the three Baltic countries. These data were taken from the EU KLEMS database.

In addition to these main variables, there are several other indicators which we use in the empirical part of the paper. The investment rate (*INV*) is measured as gross

¹³ Bulgaria and Romania are not covered in the EU KLEMS database.

¹⁴ While data on mining and quarrying (NACE C), electricity, gas and water supply (NACE E) and manufacture of leather and leather products (DC) are generally available, these sections are excluded from our sample. The reason for doing so is their high regulation (C and E) or very small share in total economy's output (DC). It has to be noted that adding these industries to our sample keeps the main results qualitatively unchanged (see Table A1 in the Annex).

¹⁵ Ideally, we would want to measure productivity as total factor productivity. Unfortunately, this and related measures are not available (or are hard to estimate in a consistent way) for the group of countries we focus on, particularly at this level of disaggregation.

¹⁶ Whenever possible, data on labour productivity and nominal value added are extrapolated to 2005 using official Eurostat sources.

¹⁷ This means that our measure of FDI inflow captures not only flow of funds, but also the revaluation effect. Unfortunately, the availability of direct data on FDI inflows is very limited, so relying on them would dramatically truncate our sample.

fixed capital formation divided by gross value added. By including the investment rate we can see whether FDI has an impact on productivity in addition to total investment. Data on gross fixed capital formation come from the Eurostat NewCronos database and are not available for Estonia, while those for Latvia and Slovenia cover only the most recent years (2003-2005 and 2000-2005, respectively).

Import penetration (*IMP*) is imports from EU-15 countries, scaled by value added. We treat this indicator as a proxy for competitive pressure. We use imports from EU-15 countries as this seems a better approximation to international competitive pressure than total imports. The data were taken from the WIIW Industrial Database Eastern Europe and are available only for manufacturing industries.

R&D intensity (*RD*) is defined as business research and development expenditures over gross value added. We include R&D expenditure as a proxy for innovation. Data on business R&D expenditure come from Eurostat and have a relatively wide industry-coverage only for the Czech Republic, Hungary and Poland.

Finally, we also use data on capacity utilisation from opinion surveys in order to capture cyclical swings in productivity. These survey data come from the European Commission's regular harmonised survey of the business sector in EU countries (see European Commission, 2007) and are available only for total manufacturing.

6. Results

6.1. Preliminary regressions

In order to establish a benchmark for our econometric choices described in Section 4, we first estimate a set of simple regressions, using only the productivity gap and FDI intensity as explanatory variables. In the case of the time series approach, we start off with simple ordinary least squares (OLS), then use the fixed effect estimator (FE) and finally run our preferred system GMM. The estimated econometric specification can be written as:¹⁸

 $\Delta \ln LP_{ijt} = \alpha_{ij} + \alpha_t + \beta \ln RLP_{ijt-1} + \gamma FDI_{ijt-1}$

¹⁸ In the OLS specification the first intercept is identical across all observations.

where subscripts i, j and t index industry, country and year, respectively, and the variables appearing in the equation are as defined in Section 5.

In the cross section approach we first run a simple OLS as a benchmark and then use our preferred SUR technique, using the following specification:¹⁹

$$\Delta \ln LP_{ijt} = \alpha_{it} + \alpha_{jt} + \beta \ln RLP_{ijt} + \gamma FDI_{ijt}$$

where subscript *t* denotes one of the two five-year subperiods. As discussed in Section 4, each SUR regression is a system of two equations, covering two five-year periods: 1996-2000 and 2001-2005.

[Table 2 about here]

The results of the preliminary regressions are reported in Table 2. As can be seen from comparing the estimates from columns 1 to 3, using OLS or FE in a dynamic panel data setup results in well-know biases of the autoregressive term. The direction of the bias is exactly as expected (see e.g. Bond, 2002): OLS clearly overpredicts the inertia in the dependent variable (and thus underpredicts the speed of convergence), while the opposite holds for the FE estimator. Comparing the results reported in columns 4 and 5 shows that adding country and industry dummies in a cross-section setup slightly changes the estimates. In particular, the coefficient of the gap term is now closer to that obtained using the time series approach.

Finally, we note that although all regressions yield significant estimates of the parameters of interest, our preferred techniques yield somewhat more conservative results in statistical terms, which is reassuring as regards the robustness of the conclusions we draw.

6.2. FDI and absorptive capacity

We start presenting our main results with a discussion of the regressions estimated using the system GMM method. The main results are reported in Table 3.

[Table 3 about here]

¹⁹ In the OLS version there is only one intercept, common across all observations of a given 5-year subperiod.

In column 1, we repeat for convenience the estimates of the simple specification considered in the previous subsection. Both the relative productivity level and the FDI share enter in a statistically significant way, pointing to a strong convergence effect and an important role of foreign capital in accounting for productivity growth in central and eastern Europe. Compared to other studies for developing countries, our estimate of the speed of convergence looks relatively high, which should not be surprising given the close integration of the central and eastern European region with the euro area. As regards the size of the coefficient on the FDI variable, those studies finding it significant usually report higher values. On the other hand, there are a number of papers that do not find any robust relationship between FDI and growth. Our estimate is both statistically and economically significant, since it implies a long-run semi-elasticity of productivity level with respect to the FDI share of about 0.9.²⁰

In column 2, we check whether FDI has effects over and above those of total investment by including investment rate as an additional explanatory variable. This may also be justified by the fact that our measure of productivity is labour productivity rather than total factor productivity. Hence, there may be some role of capital deepening in accounting for productivity developments. However, the results hardly change compared to those reported in column 1. Since including the investment rate leads to a sizable decrease in the number of observations effectively used (see Section 5), we run the remaining regressions without this control.

We next examine the effect of adding an interaction between FDI and the relative productivity level. As can be seen from column 3, we find a positive and significant coefficient, which is consistent with the absorptive capacity argument of Glass and Saggi (1998). Together with the coefficient on *FDI* without interaction becoming insignificant, this may suggest that inflow of foreign capital positively affects productivity only if the distance to the technological frontier is not too large.

In column 4, we augment the specification from column 1 by including the proxy for human capital. It turns out insignificant and does not change the remaining estimates in qualitative terms, while implying a slightly lower speed of convergence and a somewhat larger long-run effect of FDI.

²⁰ This becomes apparent once one realises that our specification can be viewed as a special case of an error-correction model.

The results reported in column 5 are obtained by adding the interaction between human capital and FDI. It turns out positive and significant, which reinforces the role of absorptive capacity in determining the impact of FDI on productivity growth and is consistent with the findings of Borensztein et al. (1998) obtained on a larger sample of developing countries.

In column 6 we replace the interaction of human capital and FDI with that of human capital and relative productivity. It does not enter in a significant way and does not lead to sizable changes in the coefficients on FDI or the relative productivity level compared to the regression reported in column 1.

Finally, in column 7 we show the results for a specification including all regressors. Unfortunately, such a specification suffers from serious econometric problems. More specifically, in the GMM estimation the number of instruments turns out too large relative to the number of observations, so the results cannot be treated as fully reliable.²¹ Nevertheless, the estimates obtained from this extended specification seem to confirm in qualitative terms all our major results discussed so far.

We repeat the six baseline steps described above using the cross section approach and employing the SUR estimation technique. The results are shown in Table 4.

[Table 4 about here]

Except for the last specification, we get a qualitatively similar picture to that of the time series approach. In particular, we find a strong convergence effect and impact of FDI inflow, the latter exhibiting patterns suggesting an important role of absorptive capacity.

An important difference emerges from comparing column 6 of Table 3 and Table 4. Using the cross section approach we find a negative and statistically significant coefficient on the interaction between human capital and the relative productivity level, while the relative productivity term becomes insignificant. This result can be interpreted as evidence for the critical role of human capital in bridging the productivity gap in central and eastern Europe, in line with the idea advocated by

²¹ In principle, one could try to alleviate this problem by truncating the number of lags in the GMMstyle instruments. Unfortunately, this is not an attractive option as tests of overidentifying restrictions and second order autocorrelation clearly reject such a simplified specification.

Nelson and Phelps (1966). However, lack of support for this finding from the time series approach suggests caution in interpreting the results this way.

6.3. The role of openness and innovation

We also estimate a set of regressions using import penetration (*IMP*) and R&D intensity (*RD*), following the same strategy as with human capital, i.e. including them alone and in interaction terms. Generally, the results are inconclusive, so we summarize them only briefly below.²²

In the system GMM regressions, import penetration alone does not enter in a statistically significant way, while its two interactions do: the one with FDI is significantly positive, while that with relative productivity is strongly negative. This might suggest that stronger competition from abroad is conducive to larger gains from foreign capital inflows and speeds up the pace of convergence at early stages of the catch-up process. However, these findings are not confirmed using the SUR technique: the interaction of import penetration with FDI has the negative sign while that with the relative productivity level is not significant. If included in the regression without interactions, import penetration turns out highly positive. All in all, although there is some evidence for the positive role of high competition in accounting for productivity growth in the countries considered in this paper, its particular channels seem rather unclear.

All regressions including R&D intensity yield insignificant coefficients on this variable and its interaction, both in the time series and the cross section approach. It has to be emphasised, however, that the coverage of the data we have on R&D is far from satisfactory.

Including R&D intensity leaves other coefficients of interest qualitatively unchanged, while the effect of adding import penetration is very similar to that of restricting the sample to manufacturing industries (see Table 5, described in the next subsection). This is not surprising as we do not have data on imports of services.

6.4. Cross-section heterogeneity

The size of our sample, although quite impressive given well known problems with data availability and comparability across transition economies, does not allow

²² Detailed results are available from the authors upon request.

us to examine cross section heterogeneity of parameters of interest using too detailed breakdowns. Still, it is feasible and potentially interesting to check how our results differ across sufficiently broadly defined groups of sectors or countries.

We do this type of exercise using only the system GMM approach for several specifications, which we consider as the key ones for the conclusions we have drawn so far. This is motivated by the fact that running SUR regressions even on two subsamples of equal size in a comparable setup to that used for the total sample, i.e. with a full set of country and industry dummies, makes the number of estimated parameters too large given the standard rule-of-thumb used in applied econometric works. Therefore, to be on the safe side, we do this exercise only for those system GMM regressions which yield qualitatively similar results to the SUR approach.

First, we examine heterogeneity in the key parameters between two groups of industries: manufacturing and services (including construction). We focus on three preferred specifications, corresponding to regressions 1, 3 and 5 in Table 3. The results are reported in Table 5.

[Table 5 about here]

Three important features stand out. One is that convergence towards euro area levels is much more pronounced in manufacturing than in services, which corroborates observations made in Section 2. Second, it is manufacturing where absorptive capacity measured as the relative productivity level is important for the positive effect of FDI inflow to materialise. Third and symmetrically, the beneficial role of FDI in services highly depends on a sufficient level of human capital.

In the second breakdown, we split our sample into two regions: Central Europe (the Czech Republic, Hungary, Poland, Slovakia and Slovenia) and the Baltic countries (Estonia, Latvia and Lithuania). Also, as documented in Section 2, the Baltic countries had relatively low initial productivity levels and embarked on the transition process towards the market economy later than the five central European countries. Since we do not have data on our proxy for human capital for any of the Baltic countries, we restrict our attention to regressions 1 and 3 from Table 3. The results are reported in Table 6.

[Table 6 about here]

It is apparent that the speed of convergence is substantially faster in the Baltic region than in the central European countries. Interestingly, in the former group, the extent of the benefits from FDI seems to depend positively on the absorptive capacity, measured as the relative productivity level vis-à-vis the euro area, while the opposite holds true for the latter countries. Putting it differently, a smaller distance to the technological frontier is accompanied by a larger positive effect from FDI in the Baltic countries, whereas in central Europe inflows of foreign capital led to particularly strong productivity gains at relatively early stages of the catching-up process. A possible explanation of this finding is that the Baltic countries had too low productivity levels in the first years of our sample to extract benefits from FDI inflows and they developed this capability only gradually.²³

6.5. Two periods of convergence

Given our findings, indicating a significant role for absorptive capacity in the convergence process, it may be interesting to examine how the relative importance of productivity determinants evolved over time. Such an exercise may be particularly useful for assessing future convergence prospects in the central and eastern European EU Member States.

Given our sample size, splitting it into two equal sub-periods makes the system GMM method rather inefficient. Therefore, this time we rely on the SUR technique and account for parameter heterogeneity over time by relaxing the restrictions on parameter equality across the equations run for the two sub-periods (1995-2000 and 2001-2005). As before, we restrict our attention to three key specifications. The results are reported in Table 7.

[Table 7 about here]

The specification including the interaction between FDI and the relative productivity level shows the most striking differences across the two sub-periods. This term is highly positive and significant in the equation estimated over the period

 $^{^{23}}$ This hypothesis seems to be confirmed by the unrestricted variant of our SUR estimations: if we allow the coefficients in regression 3 from Table 4 to vary across the two sub-periods, we get a positive and significant estimate of the interaction term only in the first equation, covering the period 1995-2000 (see Table 7).

1995-2000, while insignificant in the second half of our sample. This suggests that productivity gains of foreign capital inflows were limited by a large technological gap at the early stages of convergence, while over time this constraint ceased to play a significant role. An important implication of these results, confirmed by those obtained from a simple specification excluding the interaction, is that FDI inflows were a main driver of productivity gains in the more advanced stages of the convergence process in the central and eastern European Member States (i.e. during the second half of our sample).

On the contrary, the interaction between FDI and human capital turns out to be significant in both sub-periods. This confirms that human capital is an important factor shaping the future path of convergence in the region.

6.6. Some robustness checks

As already mentioned before, one of the weaknesses of the time series approach is that its results may be driven by cyclical rather than medium- or longterm movements. There are certainly grounds to assume that the observed procyclicality of labour productivity is to some extent due to imperfect measurement of changes in utilisation of factor inputs (see Basu and Kimball, 1997). In our case, this problem should not be very serious, since we measure labour productivity as output per hour worked rather than per person employed, hence changes in working time are explicitly taken into account. Still, it is plausible that effort per hour is not constant over the business cycle, which means that our measure of productivity may exhibit some cyclical patterns related to imperfect measurement of effective labour input.

To deal with this issue we re-estimate all regressions from Table 3 with log changes in capacity utilisation in manufacturing as a control variable.²⁴ This does not affect any of the main results obtained from the baseline specification (see Table A2 in the Annex). It has to be noted, however, that this robustness check can be treated only as a very rough one, since we do not have industry-specific measures of capacity utilisation at the level of detail in this study.

Finally, we check whether our main results are not driven by any single industry that is insignificant for the total economy. Looking at the value added

²⁴ This is the approach pursued by Cameron et al. (2005) in a similar setup covering UK manufacturing industries.

composition across industries in the central and eastern European Member States, the share of hotels and restaurants (NACE H) stands out as relatively small in all countries, while the coke and refinery industry (NACE DF) is virtually nonexistent in the Czech Republic, Estonia, Latvia, and Slovenia. Excluding these industries from our sample does not change the main results in a qualitatively significant way, however (see Table A3 in the Annex). This is the case for both the time series and the cross section approach.

7. Conclusions

The central and eastern European EU Member States have recorded impressive productivity gains over the past 15 years. At the broad sectoral level, manufacturing has been the main driver of productivity convergence, whereas gains in services have been less pronounced. Despite this catching-up process, however, a marked gap vis-àvis the rest of the EU remains. Productivity gains have been accompanied by substantial inflows of FDI, particularly to financial and business-related services and, to a lesser extent, to industry. These general trends, however, mask important differences at the country and industry level.

The empirical results in this paper point to three main conclusions, which seem to be robust to a variety of tests. First, there is a strong convergence effect in productivity both at the country and at the industry level, i.e. productivity growth depends positively on its gap vis-à-vis the euro area. At the country level, this effect is highly pronounced in the Baltic region. At the industry level, the convergence effect is particularly strong in the manufacturing sector. Second, foreign capital, in the form of FDI inflows, plays an important role in accounting for productivity growth in the central and eastern European region. Third, the impact of FDI on productivity critically depends on the absorptive capacity. More specifically, the effect of FDI on productivity seems to be increasing with a declining productivity differential vis-à-vis the euro area. There is also evidence that the level of human capital is positively associated with a larger impact of FDI. The former type of interaction between absorptive capacity and the beneficial impact from FDI seems to be important in manufacturing, whereas the latter is more significant in services.

References

- Arellano, M., Bond, S.R., 1991, Some Tests of Specification for Panel Data: Monte Carlo Evidence and an Application to Employment Equations, Review of Economic Studies 58: 277-297.
- Arellano, M., Bover, O., 1995, Another Look at the Instrumental-Variable Estimation of Error-Components Models, Journal of Econometrics 68: 29-51.
- Arratibel, O., Heinz, F.F., Martin, R., Przybyla, M., Rawdanowicz, L., Serafini, R., Zumer, T., 2007, Determinants of growth in the central and eastern European EU member states – a production function approach, ECB Occasional Paper No. 61.
- Balasubramanyam, V. N., Salisu, M., Dapsoford, D., 1999, Foreign Direct Investment as an Engine of Growth. Journal of International Trade and Economic Development 8(1): 27-40.
- Barrell, R., Holland, D., 2000, Foreign direct investment and enterprise restructuring in Central Europe, Economics of Transition 8(2): 477-504.
- Barro, R.J., 1991, Economic Growth in a Cross Section of Countries, Quarterly Journal of Economics 106: 407-443.
- Barro, R.J., Sala-i-Martin, X., 1997, Technological Diffusion, Convergence and Growth, Journal of Economic Growth 1: 1-26.
- Basu, S., Kimball, M.S., 1997, Cyclical Productivity with Unobserved Input Variation, NBER Working Paper No. 5915.
- Benhabib, J., Spiegel, M.M., 2005, Human Capital and Technology Diffusion, in: Aghion P. and Durlauf S. [eds.], Handbook of Economic Growth, Elsevier, North Holland.
- Blanchard, O., 1997, The Economics of Post-Communist Transition, Clarendon Lectures in Economics, Oxford University Press.
- Blundell, R.W., Bond, S.R., 1998, Initial Conditions and Moment Restrictions in Dynamic Panel Data Models, Journal of Econometrics 87: 115-143.
- Bond, S.R., 2002, Dynamic Panel Data Models: A Guide to Micro Data Methods and Practice, IFS Working Paper No. 09/02.
- Borensztein, E., De Gregorio, J., Lee, J., 1998, How does Foreign Direct Investment Affect Economic Growth? Journal of International Economics 45: 115-135.
- Bound, J.D., Jaeger, D.A., Baker, R.M., 1995, Problems with Instrumental Variables Estimation When the Correlation between the Instruments and Endogenous Variables Is Weak, Journal of the American Statistical Association 90: 443-450.
- Cameron, G., Proudman, J., Redding, S., 2005, Technological convergence, R&D, trade and productivity growth, European Economic Review 49: 775-807.
- Campos, N.F., Coricelli, F., 2002, Growth in Transition: What We Know, What We Don't, and What We Should, Journal of Economic Literature 40(3): 793-836.
- Carkovic, M., Levine, R., 2005, Does Foreign Direct Investment Accelerate Economic Growth? In: Moran, T.H, Graham, E.M., Blomstrom, M. [eds.], Does Foreign Direct Investment Promote Development? Peterson Institute.

- Castejón, C.F., Wörz, J., 2007, Good or Bad? The Influence of FDI on Output Growth: An industry-level analysis, WIIW Working Paper No. 38.
- Cohen, W.M., Levinthal, D.A., 1989, Innovation and Learning: The Two Faces of R&D, Economic Journal 99: 569-596.
- Damijan J.P., Knell M., Majcen B., Rojec M., 2003, The role of FDI, R&D accumulation and trade in transferring technology to transition countries: evidence from firm panel data for eight transition countries, Economic Systems 27(2): 189-204.
- Damijan, J.P., Rojec, M., 2007, Foreign Direct Investment and the Catching-up Process in New EU Member States: Is There a Flying Geese Pattern?, Applied Economics Quarterly 53(2): 91-118.
- Djankov, S., Hoekman, B.M., 2000, Foreign Investment and Productivity Growth in Czech Enterprises, World Bank Economic Review 14(1): 49-64.
- Doyle, P., Kuijs, L., Jiang, G., 2001, Real Convergence to EU Income Levels: Central Europe from 1990 to the Long Term, IMF Working Paper No. 01/146.
- Duczynski, P., 2003, Convergence in a model with technological diffusion and capital mobility, Economic Modelling 20: 729-740.
- European Bank for Reconstruction and Development, 1994, Transition Report, London.
- European Commission, 2004, The EU Economy: 2004 Review, Brussels.
- European Commission, 2007, The Joint Harmonised EU Programme of Business and Consumer Surveys User Guide (Updated 4 July 2007), Brussels.
- Findlay, R., 1978, Relative Backwardness, Direct Foreign Investment, and the Transfer of Technology: A Simple Dynamic Model, Quarterly Journal of Economics 92: 1-16.
- Gersl, A., Rubene, I., Zumer, T., 2007, Foreign Direct Investment and Productivity Spillovers: Updated Evidence from Central and Eastern Europe, Czech National Bank Working Paper No 8.
- Glass, A., Saggi, K., 2002, Multinational Firms and Technology Transfer, Scandinavian Journal of Economics 104: 495-514.
- Görg, H., Greenaway, D., 2004, Much Ado About Nothing? Do Domestic Firms Really Benefit from Foreign Direct Investment? World Bank Research Observer 19(2): 171-197.
- Gorodnichenko, Y., Svejnar, J., Terrell, K., 2007. When Does FDI Have Positive Spillovers? Evidence from 17 Emerging Market Economies, CEPR Discussion Paper No. 6546.
- Griffith, R., Redding, S., Van Reenen, J., 2004, Mapping the Two Faces of R&D: Productivity Growth in a Panel of OECD Industries, Review of Economics and Statistics 86: 883-895.
- Grossman, G.M., Helpman, E., 1991, Innovation and Growth in the Global Economy, MIT Press, Cambridge MA.

- Holland, D., Pain, N., 1998, The Diffusion of Innovations in Central and Eastern Europe: A Study of the Determinants and Impact of Foreign Direct Investment, NIESR Discussion Paper No. 137.
- Hunya, G., Large Privatisation, Restructuring and Foreign Direct Investment, in: Zecchini, S. (Ed.), Lessons from the Transition: Central and Eastern Europe in the 1990s, Dordrecht: Kluwer Academic.
- Hunya, G., Schwarzhappel, M., 2007, WIIW Database on Foreign Direct Investment in Central, East and Southeast Europe: Shift to the East, FDI05/2007, WIIW.
- Islam, N., 1995, Growth Empirics: A Panel Data Approach, Quarterly Journal of Economics 110: 1127-1170.
- Kokko, A., 1994, Technology, Market Characteristics, and Spillovers, Journal of Development Economics 43: 279-293.
- Kolasa, M., 2008, How does FDI inflow affect productivity of domestic firms? The role of horizontal and vertical spillovers, absorptive capacity and competition, Journal of International Trade and Economic Development 17(1): 155-173.
- Koszerek, D., Havik K., Mc Morrow K., Röger W. and Schönborn F., 2007, An overview of the EU KLEMS Growth and Productivity Accounts, DG-Ecfin Economic Paper No. 290.
- Lenain, P., Rawdanowicz, L., 2004, Enhancing income convergence in central Europe after EU accession, OECD Economics Department Working Paper No. 392.
- Mankiw, N.G., Romer, D., Weil, D., 1992, A Contribution to the Empirics of Economic Growth, Quarterly Journal of Economics 107: 407-437.
- Mencinger, J., 2003, Does Foreign Direct Investment Always Enhance Economic Growth?, Kyklos 56(4): 491-508.
- Nelson, R.R., Phelps, E.S., 1966, Investment in Humans, Technological Diffusion, and Economic Growth, American Economic Review 56: 69-75.
- Nelson, C.R., Startz, R., 1990, The Distribution of the Instrumental Variables Estimator and Its t-Ratio When the Instrument is a Poor One, Journal of Business 63: 125-140.
- Nickell, S., 1981, Biases in Dynamic Models with Fixed Effects, Econometrica 49: 1417-1426.
- Romer, P.M., 1990, Endogenous Technological Change, Journal of Political Economy 98: 71-102.
- Romer, P., 1993, Idea gaps and object gaps in economic development, Journal of Monetary Economics 32(3).
- Roodman, D., 2006, How to Do xtabond2: An Introduction to "Difference" and "System" GMM in Stata, Working Paper No. 103, Center for Global Development, Washington.
- Schadler, S., Ashoka, M., Abiad, A., Leigh, D., 2006, Growth in the Central and Eastern European Countries of the European Union, Occasional Paper No. 252, IMF, Washington, DC.

- Timmer, M., O'Mahony, M., van Ark, B., 2007, The EU KLEMS Growth and Productivity Accounts: An Overview, University of Groningen & University of Birmingham.
- Wang, J-Y., 1990, Growth, technology transfer, and the long-run theory of international capital movements, Journal of International Economics 29: 255-271.
- World Bank, 2008, Global Economic Prospects Technology Diffusion in the Developing World, Washington.



Chart 1 Labour productivity levels relative to the euro area (euro area = 100)

50

40

30

20

10

0

CZ

EE

HU

LV

LT

■ 1995 ■ 1999 ■ 2003 ■ 2006

PL

SK

SI

Source: EU KLEMS and Eurostat. Labour productivity is calculated as gross value added per hour worked. Level comparisons based on industry-specific PPPs.

50

40

30

20

10

0

CZ

EE

HU

LV

LT

□ 1995 □ 1999 □ 2003 □ 2006

PL

SK

SI



Source: EU KLEMS and Eurostat. CEE-8: CZ, EE, HU, LV, LT, PL, SK, SI.



Source: WIIW and Eurostat. Hungary: 1998 instead of 1997.



Source: WIIW, EU KLEMS and Eurostat. CEE-8: CZ, EE, HU, LV, LT, PL, SK, SI. Note: Country composition changes due to differences in data availability (see country charts in the Annex).

Variable	Definition	Source	Availability
Labour productivity (LP)	Value added per hour worked	EU KLEMS	Total sample
Relative labour productivity (RLP)	Labour productivity level relative to the euro area	EU KLEMS	Total sample
Foreign Direct Investment (FDI)	Gross FDI inflow (calculated from the change in stocks) as a share of value added	WIIW database on Foreign Direct Investment in Central, East and Southeast Europe	Available for 1995- 2005 for LT, LV, PL and SI, while from 1996(7) for the other countries
Total investment (INV)	Gross fixed capital formation as a share of value added	Eurostat (NewCronos)	Not available for EE, gaps for LV and SI
Human capital (HC)	Share of high-skilled workers in total hours worked	EU KLEMS	Not available for EE, LT and LV.
Import penetration (IMP)	Imports from EU-15 as a share of value added	WIIW Industrial Database Eastern Europe	Available for manufacturing only
R&D intensity (RD)	Business R&D expenditure as a share of value added	Eurostat	Wide coverage only for CZ, HU and PL
Capacity utilisation	Producers' assessment of the current level of capacity utilisation	European Commission Industry Survey	Available for manufacturing only

Table 1. Data – Definitions of variables

Note: Relative labour productivity levels vis-à-vis the euro area for 1997 are calculated using industry-specific purchasing power parities (PPPs). Estimates for the remaining years are extrapolated using labour productivity growth rates.



$\Delta \ln LP$	(1)	(2)	(3)	(4)	(5)
ln <i>RLP</i>	-0.057***	-0.518***	-0.148**	-0.045***	-0.064***
	(0.008)	(0.026)	(0.064)	(0.006)	(0.010)
FDI	0.121***	0.064**	0.129**	0.117***	0.098***
	(0.026)	(0.027)	(0.056)	(0.027)	(0.032)
Estimation Method	OLS	FE	GMM	OLS	SUR
Observations	1075	1075	1075	294	294

Table 2. Preliminary regressions

Notes: The estimations are performed using the ordinary least squares (OLS), fixed effects with a full set of country-industry dummies (FE), the system generalized method of moments (GMM) and the seemingly unrelated technique (SUR); for details on GMM and SUR estimations, see notes to Table 3 and Table 4, respectively; for the time-series approach (columns 1 to 3), the sample is an unbalanced panel of yearly observations covering the period of 1996-2005; for the cross-section approach (columns 4 and 5), each regression is a system of two equations, covering two five-year periods: 1996-2000 and 2001-2005; *LP* is labour productivity; *RLP* is labour productivity relative to that of the euro area; *FDI* is the foreign direct investment share in value added; numbers in parentheses are robust standard errors; *, ** and *** denote 10%, 5% and 1% statistical significance, respectively.

$\Delta \ln LP$	(1)	(2)	(3)	(4)	(5)	(6)	(7)
ln <i>RLP</i>	-0.148** (0.064)	-0.138*** (0.044)	-0.071** (0.032)	-0.098* (0.056)	-0.099* (0.053)	-0.152** (0.076)	-0.174** (0.074)
FDI	0.129** (0.056)	0.127*** (0.042)	-0.360 (0.282)	0.133*** (0.049)	0.050 (0.059)	0.152*** (0.055)	0.921*** (0.332)
INV		-0.006 (0.039)					-0.082 (0.053)
FDI · lnRLP			0.144* (0.079)				0.241*** (0.088)
НС				-0.027 (0.125)	-0.081 (0.141)	-2.695 (1.788)	-3.391** (1.645)
$HC \cdot \ln RLP$						0.654 (0.448)	0.820** (0.407)
HC · FDI					0.479* (0.279)		0.670** (0.263)
Observations	1075	804	1075	617	617	617	534
Serial correlation test (p-value)	0.32	0.59	0.33	0.12	0.15	0.13	0.04
Hansen test (p-value)	0.33	0.46	0.25	0.32	0.38	0.45	0.90

Table 3. System GMM estimation results

Notes: The estimation is done using the system GMM technique; the sample is an unbalanced panel of yearly observations covering the period of 1996-2005; all regressions include a full set of time dummies; *LP* is labour productivity; *RLP* is labour productivity relative to that of the euro area; *FDI* is the foreign direct investment share in value added; *INV* is the gross fixed capital formation share in value added; *HC* is the share of high-skilled workers in employment; all regressors are lagged one year; numbers in parentheses are robust standard errors; the null hypothesis of the serial correlation test is that errors in the first differenced regression exhibit no second order correlation; the null hypothesis of the Hansen test is that the instruments are exogenous; *, ** and *** denote 10%, 5% and 1% statistical significance, respectively.

$\Delta \ln LP$	(1)	(2)	(3)	(4)	(5)	(6)
ln <i>RLP</i>	-0.064*** (0.010)	-0.056*** (0.012)	-0.068*** (0.010)	-0.045*** (0.012)	-0.050*** (0.014)	-0.001 (0.020)
FDI	0.098*** (0.032)	0.159*** (0.058)	-0.093 (0.117)	0.190*** (0.055)	-0.004 (0.098)	0.105* (0.059)
INV		-0.020 (0.034)				
FDI · lnRLP			0.057* (0.034)			
НС				-0.306 (0.314)	-0.590* (0.350)	1.149** (0.531)
$HC \cdot \ln RLP$						-0.372*** (0.108)
HC · FDI					1.133** (0.471)	
Observations R^2 for individual periods	294 0.61, 0.33	232 0.67, 0.38	294 0.62, 0.32	172 0.66, 0.45	172 0.66, 0.36	172 0.67, 0.43

Table 4. SUR estimation results

Notes: The estimation is done using the SUR technique; each regression is a system of two equations, covering two five-year periods: 1996-2000 and 2001-2005; the estimation allows for different error variances in each equation and for correlation of these errors across equations; all regressions include a full set of country and industry dummies, the coefficients on which are allowed to vary across periods; other coefficients are constrained to be the same for both periods; LP is labour productivity; RLP is labour productivity relative to that of the euro area; FDI is the foreign direct investment share in value added; INV is the gross fixed capital formation share in value added; HC is the share of high-skilled workers in employment; all variables are expressed as five-year averages, except for RLP, which is measured for the year preceding the beginning of the relevant period; numbers in parentheses are robust standard errors; *, ** and *** denote 10%, 5% and 1% statistical significance, respectively.

$\Delta \ln LP$	(1m)	(1s)	(3m)	(3s)	(5m)	(5s)
ln <i>RLP</i>	-0.283** (0.113)	-0.066 (0.050)	-0.112** (0.045)	-0.051 (0.045)	-0.181* (0.100)	-0.028 (0.058)
FDI	0.123 (0.092)	0.129*** (0.039)	-0.543** (0.269)	0.263 (0.519)	0.214 (0.239)	0.060 (0.114)
FDI · lnRLP			0.223*** (0.086)	-0.035 (0.134)		
НС					-0.222 (0.411)	-0.085 (0.108)
HC · FDI					-1.681 (2.711)	1.464*** (0.386)
Observations	685	390	685	390	386	231

Table 5. System GMM estimation results - manufacturing vs. services

Notes: See notes to Table 3. The column numbers correspond to the relevant regressions in Table 3, with 'm' denoting manufacturing (NACE DA to DN, without DC), while 's' stands for services (including construction, NACE F to K).

$\Delta \ln LP$	(1CE)	(1BL)	(3CE)	(3BL)
ln <i>RLP</i>	-0.131* (0.068)	-0.226*** (0.077)	-0.082 (0.072)	-0.151** (0.061)
FDI	0.127*** (0.041)	0.186* (0.107)	1.186** (0.516)	-0.559*** (0.193)
FDI · lnRLP			-0.288** (0.137)	0.227*** (0.072)
Observations	662	413	662	413

Table 6. System GMM estimation results - CEE vs. Baltic countries

Notes: See notes to Table 3. The column numbers correspond to the relevant regressions from Table 3, with the following acronyms used for the two regions: CE (Central Europe: the Czech Republic, Hungary, Poland, Slovakia, Slovenia), BL (Baltic countries: Estonia, Latvia, Lithuania).

$\Delta \ln LP$	(1_95-00)	(1_01-05)	(3_95-00)	(3_01-05)	(5_95-00)	(5_01-05)
ln <i>RLP</i>	-0.069*** (0.013)	-0.046*** (0.013)	-0.071*** (0.013)	-0.042*** (0.013)	-0.050** (0.100)	-0.035** (0.015)
FDI	0.063 (0.044)	0.114*** (0.038)	-0.410** (0.173)	0.430** (0.215)	-0.019 (0.140)	-0.036 (0.123)
$FDI \cdot \ln RLP$			0.155*** (0.056)	-0.080 (0.056)		
НС					-0.431 (0.438)	-0.669* (0.368)
HC · FDI					1.614* (0.937)	1.155** (0.544)
Observations	147	147	147	147	86	86

 Table 7. Unrestricted SUR estimation results

Notes: The difference compared to the regressions reported in Table 4 is that the estimation does not restrict any of the parameters to be equal across the two subperiods. Otherwise, see notes to Table 4. The column numbers correspond to the relevant regressions from Table 4, with '_95-00' denoting the first five-year period (1995-2000), while '_01-05' standing for the second period (2001-2005).

ANNEX



Chart A1. Cumulative labour productivity growth by country (1995 = 100)

Source: EU KLEMS and Eurostat.





PM: Value for Estonia in 2004: 210.

Chart A2. FDI stocks as a share of value added by country (in %)

Source: WIIW, EU KLEMS and Eurostat.



$\Delta \ln LP$	(1_GMM)	(1_SUR)	(3_GMM)	(3_SUR)	(5_GMM)	(5_SUR)
ln <i>RLP</i>	-0.140*** (0.115)	-0.064*** (0.009)	-0.092*** (0.033)	-0.068*** (0.009)	-0.147** (0.063)	-0.048*** (0.012)
FDI	0.115** (0.053)	0.099*** (0.030)	-0.488** (0.237)	-0.119 (0.114)	0.081 (0.067)	-0.015 (0.085)
$FDI \cdot \ln RLP$			0.181** (0.071)	0.066** (0.033)		
НС					0.012 (0.154)	-0.137 (0.317)
HC · FDI					0.424* (0.252)	1.138*** (0.428)
Observations	1223	340	1223	340	672	192

Table A1. Main results – industries C, E and DC included

Notes: The column numbers correspond to the relevant regressions from Table 3 (GMM) and Table 4 (SUR), see notes to these tables. Compared to the baseline, the sample also includes the following industries: mining and quarrying (NACE C), electricity, gas and water supply (NACE E) and manufacture of leather and leather products (DC).

$\Delta \ln LP$	(1)	(3)	(5)
ln <i>RLP</i>	-0.165** (0.067)	-0.105*** (0.040)	-0.113** (0.053)
FDI	0.133** (0.057)	-0.380 (0.264)	0.046 (0.060)
$FDI \cdot \ln RLP$		0.151** (0.075)	
НС			-0.069 (0.142)
HC · FDI			0.489* (0.276)
Observations	1075	1075	617

Table A2. Main results - capacity utilization included

Notes: The column numbers correspond to the relevant regressions from Table 3, see notes to this table. Compared to the baseline, capacity utilization is included as an additional regressor (not reported).

$\Delta \ln LP$	(1_GMM)	(1_SUR)	(3_GMM)	(3_SUR)	(5_GMM)	(5_SUR)
ln <i>RLP</i>	-0.136*** (0.048)	-0.046*** (0.009)	-0.066* (0.034)	-0.044*** (0.010)	-0.171*** (0.064)	-0.063*** (0.014)
FDI	0.101** (0.050)	0.118*** (0.027)	-0.465** (0.192)	-0.017 (0.038)	0.025 (0.069)	-0.010 (0.090)
FDI · lnRLP			0.165*** (0.055)	0.176* (0.137)		
НС					0.055 (0.177)	-0.424 (0.371)
HC · FDI					0.475* (0.276)	1.078** (0.439)
Observations	993	274	993	274	564	160

Table A3. Main results - industries H and DF excluded

Notes: The column numbers correspond to the relevant regressions from Table 3 (GMM) and Table 4 (SUR), see notes to these tables. Compared to the baseline, the sample excludes the following industries: hotels and restaurants (NACE H) and coke and refinery (DF).

European Central Bank Working Paper Series

For a complete list of Working Papers published by the ECB, please visit the ECB's website (http://www.ecb.europa.eu).

- 944 "The New Area-Wide Model of the euro area: a micro-founded open-economy model for forecasting and policy analysis" by K. Christoffel, G. Coenen and A. Warne, October 2008.
- 945 "Wage and price dynamics in Portugal" by C. Robalo Marques, October 2008.
- 946 "Macroeconomic adjustment to monetary union" by G. Fagan and V. Gaspar, October 2008.
- 947 "Foreign-currency bonds: currency choice and the role of uncovered and covered interest parity" by M. M. Habib and M. Joy, October 2008.
- 948 "Clustering techniques applied to outlier detection of financial market series using a moving window filtering algorithm" by J. M. Puigvert Gutiérrez and J. Fortiana Gregori, October 2008.
- 949 "Short-term forecasts of euro area GDP growth" by E. Angelini, G. Camba-Méndez, D. Giannone, L. Reichlin and G. Rünstler, October 2008.
- 950 "Is forecasting with large models informative? Assessing the role of judgement in macroeconomic forecasts" by R. Mestre and P. McAdam, October 2008.
- 951 "Exchange rate pass-through in the global economy: the role of emerging market economies" by M. Bussière and T. Peltonen, October 2008.
- 952 "How successful is the G7 in managing exchange rates?" by M. Fratzscher, October 2008.
- 953 "Estimating and forecasting the euro area monthly national accounts from a dynamic factor model" by E. Angelini, M. Bańbura and G. Rünstler, October 2008.
- 954 "Fiscal policy responsiveness, persistence and discretion" by A. Afonso, L. Agnello and D. Furceri, October 2008.
- 955 "Monetary policy and stock market boom-bust cycles" by L. Christiano, C. Ilut, R. Motto and M. Rostagno, October 2008.
- 956 "The political economy under monetary union: has the euro made a difference?" by M. Fratzscher and L. Stracca, November 2008.
- 957 "Modeling autoregressive conditional skewness and kurtosis with multi-quantile CAViaR" by H. White, T.-H. Kim, and S. Manganelli, November 2008.
- 958 "Oil exporters: in search of an external anchor" by M. M. Habib and J. Stráský, November 2008.
- 959 "What drives U.S. current account fluctuations?" by A. Barnett and R. Straub, November 2008.
- 960 "On implications of micro price data for macro models" by B. Maćkowiak and F. Smets, November 2008.
- 961 "Budgetary and external imbalances relationship: a panel data diagnostic" by A. Afonso and C. Rault, November 2008.
- 962 "Optimal monetary policy and the transmission of oil-supply shocks to the euro area under rational expectations" by S. Adjemian and M. Darracq Pariès, November 2008.

- 963 "Public and private sector wages: co-movement and causality" by A. Lamo, J. J. Pérez and L. Schuknecht, November 2008.
- 964 "Do firms provide wage insurance against shocks? Evidence from Hungary" by G. Kátay, November 2008.
- 965 "IMF lending and geopolitics" by J. Reynaud and J. Vauday, November 2008.
- 966 "Large Bayesian VARs" by M. Bańbura, D. Giannone and L. Reichlin, November 2008.
- 967 "Central bank misperceptions and the role of money in interest rate rules" by V. Wieland and G. W. Beck, November 2008.
- 968 "A value at risk analysis of credit default swaps" by B. Raunig and M. Scheicher, November 2008.
- 969 "Comparing and evaluating Bayesian predictive distributions of asset returns" by J. Geweke and G. Amisano, November 2008.
- 970 "Responses to monetary policy shocks in the east and west of Europe" by M. Jarociński, November 2008.
- 971 "Interactions between private and public sector wages" by A. Afonso and P. Gomes, November 2008.
- 972 "Monetary policy and housing prices in an estimated DSGE for the US and the euro area" by M. Darracq Pariès and A. Notarpietro, November 2008.
- 973 "Do China and oil exporters influence major currency configurations?" by M. Fratzscher and A. Mehl, December 2008.
- 974 "Institutional features of wage bargaining in 23 European countries, the US and Japan" by P. Du Caju, E. Gautier, D. Momferatou and M. Ward-Warmedinger, December 2008.
- 975 "Early estimates of euro area real GDP growth: a bottom up approach from the production side" by E. Hahn and F. Skudelny, December 2008.
- 976 "The term structure of interest rates across frequencies" by K. Assenmacher-Wesche and S. Gerlach, December 2008.
- 977 "Predictions of short-term rates and the expectations hypothesis of the term structure of interest rates" by M. Guidolin and D. L. Thornton, December 2008.
- 978 "Measuring monetary policy expectations from financial market instruments" by M. Joyce, J. Relleen and S. Sorensen, December 2008.
- 979 "Futures contract rates as monetary policy forecasts" by G. Ferrero and A. Nobili, December 2008.
- 980 "Extracting market expectations from yield curves augmented by money market interest rates: the case of Japan" by T. Nagano and N. Baba, December 2008.
- 981 "Why the effective price for money exceeds the policy rate in the ECB tenders?" by T. Välimäki, December 2008.
- 982 "Modelling short-term interest rate spreads in the euro money market" by N. Cassola and C. Morana, December 2008.
- 983 "What explains the spread between the euro overnight rate and the ECB's policy rate?" by T. Linzert and S. Schmidt, December 2008.

- 984 "The daily and policy-relevant liquidity effects" by D. L. Thornton, December 2008.
- 985 "Portuguese banks in the euro area market for daily funds" by L. Farinha and V. Gaspar, December 2008.
- 986 "The topology of the federal funds market" by M. L. Bech and E. Atalay, December 2008.
- 987 "Probability of informed trading on the euro overnight market rate: an update" by J. Idier and S. Nardelli, December 2008.
- 988 "The interday and intraday patterns of the overnight market: evidence from an electronic platform" by R. Beaupain and A. Durré, December 2008.
- 989 "Modelling loans to non-financial corporations in the euro area" by C. Kok Sørensen, D. Marqués Ibáñez and C. Rossi, January 2009.
- 990 "Fiscal policy, housing and stock prices" by A. Afonso and R. M. Sousa, January 2009.
- 991 "The macroeconomic effects of fiscal policy" by A. Afonso and R. M. Sousa, January 2009.
- 992 "FDI and productivity convergence in central and eastern Europe: an industry-level investigation" by M. Bijsterbosch and M. Kolasa, January 2009.